

# G OPEN ACCESS

**Citation:** Edwards AC, Larsson Lönn S, Chartier KG, Lannoy S, Sundquist J, Kendler KS, et al. (2024) Socioeconomic position indicators and risk of alcohol-related medical conditions: A national cohort study from Sweden. PLoS Med 21(3): e1004359. https://doi.org/10.1371/journal. pmed.1004359

Received: November 2, 2023

Accepted: February 14, 2024

Published: March 19, 2024

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1004359

**Copyright:** © 2024 Edwards et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be shared publicly because they are only available through application to Statistics Sweden (http:// www.scb.se/) for researchers who meet the criteria for access. **RESEARCH ARTICLE** 

# Socioeconomic position indicators and risk of alcohol-related medical conditions: A national cohort study from Sweden

Alexis C. Edwards<sup>1</sup>\*, Sara Larsson Lönn<sup>2</sup>, Karen G. Chartier<sup>1,3</sup>, Séverine Lannoy<sup>1</sup>, Jan Sundquist<sup>2,4,5</sup>, Kenneth S. Kendler<sup>1†</sup>, Kristina Sundquist<sup>2,4,5†</sup>

1 Virginia Institute for Psychiatric and Behavioral Genetics, Department of Psychiatry, Virginia Commonwealth University, Richmond, Virginia, United States of America, 2 Center for Primary Health Care Research, Lund University, Malmö, Sweden, 3 School of Social Work, Virginia Commonwealth University, Richmond, Virginia, United States of America, 4 Department of Family Medicine and Community Health, Icahn School of Medicine at Mount Sinai, New York, New York, United States of America, 5 Department of Population Health Science and Policy, Icahn School of Medicine at Mount Sinai, New York, New York, United States of America

‡ These authors share last authorship on this work.

# Abstract

# Background

Alcohol consumption contributes to excess morbidity and mortality in part through the development of alcohol-related medical conditions (AMCs, including alcoholic cardiomyopathy, hepatitis, cirrhosis, etc.). The current study aimed to clarify the extent to which risk for these outcomes differs as a function of socioeconomic position (SEP), as discrepancies could lead to exacerbated health disparities.

# Methods and findings

We used longitudinal Swedish national registries to estimate the individual and joint associations between 2 SEP indicators, educational attainment and income level, and risk of AMC based on International Classification of Diseases codes, while controlling for other sociode-mographic covariates and psychiatric illness. We conducted Cox proportional hazards models in sex-stratified analyses (N = 1,162,679 females and N = 1,196,659 males), beginning observation at age 40 with follow-up through December 2018, death, or emigration. By the end of follow-up, 4,253 (0.37%) females and 11,183 (0.93%) males had received an AMC registration, corresponding to overall AMC incidence rates among females and males of 2.01 and 5.20, respectively. In sex-stratified models adjusted for birth year, marital status, region of origin, internalizing and externalizing disorder registrations, and alcohol use disorder (AUD) registration, lower educational attainment was associated with higher risk of AMC in both females (hazard ratios [HRs] = 1.40 to 2.46 for low- and mid-level educational attainment across 0 to 15 years of observation) and males (HRs = 1.13 to 1.48). Likewise, risk of AMC was increased for those with lower income levels (females: HRs = 1.10 to 5.86; males: HRs = 1.07 to 6.41). In secondary analyses, we further adjusted for aggregate familial risk of

<sup>\*</sup> alexis.edwards@vcuhealth.org

**Funding:** This project was supported by grant AA023534 from the US National Institutes of Health to KK and KS, and grants from the Swedish Research Council to JS (2020-01175) as well as ALF funding from Region Skåne awarded to KS. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

**Abbreviations:** AD, anxiety disorder; AMC, alcoholrelated medical condition; ATC, Anatomical Therapeutic Chemical; AUD, alcohol use disorder; DUD, drug use disorder; ED, externalizing disorder; HR, hazard ratio; ID, internalizing disorder; MD, major depression; RERI, relative excess risk due to interaction; SEP, socioeconomic position. AUD by including family genetic risk scores for AUD (FGRS<sub>AUD</sub>), estimated using medical, pharmacy, and criminal registries in extended families, as covariates. While FGRS<sub>AUD</sub> were associated with risk of AMC in adjusted models (HR = 1.17 for females and HR = 1.21 for males), estimates for education and income level remained largely unchanged. Furthermore, FGRS<sub>AUD</sub> interacted with income level, but not education level, such that those at higher familial liability to AUD were more susceptible to the adverse effect of low income. Limitations of these analyses include the possibility of false negatives for psychiatric illness registrations, changes in income after age 40 that were not accounted for due to modeling restrictions, restriction to residents of a high-income country, and the inability to account for individual-level alcohol consumption using registry data.

### Conclusions

Using comprehensive national registry data, these analyses demonstrate that individuals with lower levels of education and/or income are at higher risk of developing AMC. These associations persist even when accounting for a range of sociodemographic, psychiatric, and familial risk factors. Differences in risk could contribute to further health disparities, potentially warranting increased screening and prevention efforts in clinical and public health settings.

#### Author summary

#### Why was this study done?

- Alcohol consumption contributes to worldwide excess morbidity and mortality, including conditions directly related to alcohol (alcohol-related medical conditions, AMCs) such as alcoholic liver disease.
- The extent to which socioeconomic position (SEP) is longitudinally associated with AMC risk is unclear, but important to understand as it may contribute to health disparities.

#### What did the researchers do and find?

- We employed a model that follows people over time to estimate their risk of AMC as a function of 2 SEP indicators: educational attainment and income.
- The model adjusted for other factors that might also be related to risk of AMC, such as marital status, history of psychiatric illness, and aggregate genetic liability to alcohol use disorder (AUD).
- We found that lower educational attainment and lower income were both associated with higher risk of AMC, even after accounting for other predictors.
- Individuals with higher genetic liability to AUD were at increased risk for AMC; these individuals were also more vulnerable to the negative effects of low income.

#### What do these findings mean?

- These results indicate that individuals with lower levels of educational attainment or lower incomes might warrant particular clinical attention with respect to screening or evaluation of alcohol consumption, as their risk of AMC is elevated. These differences in risk could exacerbate poor health outcomes among people with lower SEP.
- Clinicians should also be aware that individuals with a family history of AUD are at higher risk of alcohol-related medical morbidities.
- Limitations include the inability to account for individual differences in alcohol consumption, potential for excluding less severe AUD cases, and inability to extend findings to less wealthy countries.

# Introduction

Excessive alcohol use and alcohol use disorder (AUD) contribute substantially to morbidity and mortality worldwide. The World Health Organization estimates that harmful alcohol use accounts for 5.1% of the global burden of disease and injury, resulting in 3 million worldwide deaths per year [1]. In the United States, 9.8% of deaths among working-age adults were attributable to excessive alcohol use [2]. In Sweden, the 2016 prevalence of heavy episodic drinking was 28.0% (12.4% in females and 43.7% in males), slightly lower than in the European Union overall; furthermore, alcohol accounted for 17.6% of unintentional and 21.1% of intentional injuries [3]. The social and psychological consequences of problematic use are considerable, with AUD being linked to higher rates of divorce [4], criminality [5], depression [6], and suicidal behavior [7,8]. Furthermore, excessive alcohol use exacts a steep economic toll: In 2010, the cost of excessive drinking in the United States was estimated at \$249 billion [9]. Importantly, alcohol-related medical conditions (AMCs) constituted a substantial proportion of the economic burden of alcohol use, with 11.4% of the 2010 costs attributed to healthcare [9].

Socioeconomic indicators are frequently correlated with alcohol outcomes, including frequency of use, AUD, and alcohol-related morbidity and mortality, and these associations exist within the context of a complex ecosystem of influences [10]. While some studies indicate that drinking frequency and quantity are higher among those with higher socioeconomic position (SEP) [10], others have found that high-frequency/low-quantity drinking is common among those with high SEP, whereas individuals of low SEP drink less frequently but at higher quantities [11–14]. Multiple indicators of lower SEP were associated with increased risk of alcoholrelated mortality in a meta-analysis [15], and van Oers [11] reported that lower levels of educational attainment were associated with alcohol-related problems in the Netherlands. In the Swedish Public Health Cohort, lower occupational class was associated with higher risk of alcohol-related health problems, and the effect of heavy episodic drinking was more pronounced among those of lower occupational class [16]. Thus, even when controlling for particularly unhealthy drinking patterns, the extant literature generally suggests that individuals with lower SEP are more likely to suffer severe alcohol-related outcomes.

While many studies report cross-sectional associations between SEP indicators and alcohol outcomes, far fewer have examined longitudinal associations. Clarifying longitudinal associations is particularly important given the possibility of bidirectional relationships [10]: Not only might low SEP lead to poor alcohol outcomes, but alcohol problems might lead to negative

economic consequences [17]. Understanding directionality can be informative to prevention and intervention efforts, i.e., if alcohol-related harm-reduction is the goal, it is necessary to confirm that low SEP is prospectively associated with adverse alcohol outcomes to determine whether targeted prevention could be effective within this group.

In the current study, we aimed to clarify the association between SEP and AMC through the use of longitudinal, nationwide Swedish registry data and Cox proportional hazards models. We examined the associations between 2 SEP indicators—namely, educational attainment and income level—and subsequent risk of developing AMC. We sought to substantively contribute to the existing literature on this topic through the use of large, minimally biased, representative data, which also allows for stratification by sex. We account for a range of potentially important socio-demographic and psychological covariates and confounders. Furthermore, we evaluate whether associations between SEP indicators and AMC shift across adulthood. We hypothesized that lower SEP indicators would be associated with higher risk of AMC and that these associations would be attenuated but persist after accounting for covariates and potential confounders.

#### Materials and methods

#### **Ethics statement**

All analyses were approved by the Regional Ethical Review Board of Lund University (no. 2008/409 and later amendments).

#### Sample

We used several Swedish nationwide registers, linked using the unique 10-digit identification number assigned at birth or immigration to all Swedish residents. The identification number was replaced by a serial number to ensure anonymity. The registry resources are described in S1 Methods, and included the population register, multi-generation register, multiple health care registers, and the Longitudinal Integrated Database for Health Insurance and Labour Market Statistics (LISA), which included information on income and education. To maximize registry coverage based on preliminary analyses, we included females and males, born between 1950 and 1970, and residing in Sweden at age 40, without a prior AMC registration. Due to missing data on education, N = 916 females and N = 845 males from the cohort were excluded; due to missing data on marital status, N = 15 females and N = 11 males were excluded.

#### Measures

Our outcome variable, AMC, was defined from Swedish medical registers by the following ICD codes, described further in <u>S1 Methods</u>: ICD8: 571.0, ICD9: 357F, 425F, 535D, 571A, 571B, 571C, 571D; and ICD10: E24.4, G31.2, G62.1, G72.1, I42.6, K29.2, K700, K701, K702, K703, K704, K709, K85.2, K86.0, and O35.4. Individuals with AMC registrations prior to age 40 were excluded.

The primary independent variables of interest were education and familial income. Income was assessed at age 40 and categorized based on the income quartiles for the working-age population in Sweden (ages 20 to 65), although analyses included both employed and unemployed individuals. Educational attainment was categorized into low (compulsory school only), mid (upper secondary school), and high (university level), based on the subject's highest level of school completion. These variables allow us to jointly account for, and distinguish the impact of, indices of individual-level SEP (educational level) and access to resources at the family level (familial income). An individual could have low educational attainment, and/or no personal income, but still have access to resources through a spouse's income.

Sociodemographic covariates included year of birth, marital status, and region of origin. Marital status was assessed at age 40 and obtained from the total population register. The categories were: unmarried, married, divorced, or widowed. Region of origin was defined by country of birth and categorized into: Finland, Western Countries, Eastern Europe, Latin America and the Caribbean, Middle East and Northern Africa, Africa (excluding Northern Africa), and Asia (excluding Middle East) and Oceania. Finland was examined separately because they represent one of the largest immigrant groups in Sweden and because they have higher AUD rates.

From Swedish medical registers, AUD was defined by the following ICD codes: ICD8: 291, 303; ICD9: 305A, 291, 303; and ICD 10: ICD-10 codes: F10.1, F10.2, F10.3, F10.4, F10.5, F10.6, F10.7, F10.8, and F10.9. Additionally, AUD registrations were identified using the Prescribed Drug Register by retrieving "Drugs used in alcohol dependence" (Anatomical Therapeutic Chemical [ATC] Classification System code N07BB); this includes disulfiram (N07BB01), acamprosate (N07BB03), naltrexone (N07BB04), or nalmefene (N07BB05). Finally, we identified AUD cases as those convicted for or suspected of at least 2 alcohol-related crimes according to law 1951:649, paragraph 4 and 4A and law 1994:1009, Chapter 20, paragraphs 4 and 5 from the Swedish Crime Register, and code 3005 and 3201 in the Suspicion register. Most of these crime types include driving or operating a boat while intoxicated. Among criminal AUD registrations, 99.7% were related to drunk driving; however, only 18.3% of AUD registrations are detected using only in the criminal registries.

Internalizing disorder (ID) was defined using the ICD codes for major depression (MD); ICD-9: 296B, 298A, 300E; and ICD-10: F32 and F33, and anxiety disorder (AD); ICD-10: F41. Externalizing disorder (ED) was defined as criminal behavior including white collar crime, property crime, or violent crime as described in detail elsewhere [18]; or drug use disorder (DUD). DUD was identified from medical registers using codes from ICD-9: 292, 304, and 305C - 305I; ICD-10: F11-F16 and from the Prescribed Drug Register if a person had retrieved (on average) more than 4 defined daily doses a day for 12 months of a drug to treat drug dependence (excluding those suffering from cancer), defined by the following ATC codes: N02A, N05C, N05BA, or N07BC. We further defined DUD from the Suspicion Register by codes 3070, 5010, 5011, and 5012; and the Crime Register if convicted by law 1968:64, paragraph 1, point 6 or and law 1951:649, paragraph 4, subsection 2 and paragraph 4A, subsection 2.

Finally, aggregate genetic liability for AUD, which was used in a series of secondary analyses, was operationalized using the family genetic risk score for AUD (FGRS<sub>AUD</sub>), which assesses genetic risk by an examination of first through fifth degree relatives, correcting for cohabitation effects which could make relatives more similar in their AUD risk. This measure is a relative measure which is operationalized as a continuous, standardized indicator of aggregate genetic risk, so that 1 unit corresponds to 1 standard deviation in the population. The details of this measure have been described in detail elsewhere [19]; FGRS have been validated in subsequent research [20].

#### Statistical analyses

These analyses were not prospectively described in an analysis plan. S1 Methods provides information on the timing of analyses. Cox proportional hazard models were utilized to estimate the time to AMC, censoring at end of follow-up (December 2018), death, or emigration. Models were stratified by sex to facilitate direct comparisons and enhance data transparency [21].

We conducted a series of preliminary analyses to decide whether time-varying coefficients were necessary for the socioeconomic predictors. This was investigated by including a linear interaction between follow-up time and education or income. If the interaction was significant and the modification of the magnitude of the hazard ratio (HR) was potentially meaningful, we elected to allow for a time-dependent coefficient in the final models.

We then pursued model building to estimate the association between education and/or income with AMC and investigated whether observed associations were robust to the inclusion of ID, ED, and AUD, which were included as time-varying variables. These models included birth year, marital status, and region of birth as covariates. In Models 1A and 1B, we tested the association between education or income, respectively, with AMC, otherwise including only the sociodemographic covariates. In Model 2, we jointly estimated these associations, including both education and income. In Model 3, we added registrations for ID and ED, as these could act as important mediators between education/income and AMC. Finally, in Model 4, to provide insight into whether education/income is associated with risk for AMC above and beyond variance that could be accounted for by the association between socioeconomic indicators and AUD, we also included AUD as a time-varying variable.

In a secondary set of analyses, we included genetic risk of AUD, operationalized as  $FGRS_{AUD}$ , to understand how genetic liability to AUD might influence the associations with the investigated risk factors.  $FGRS_{AUD}$  is estimated with greater precision for Swedish born subjects with 2 Swedish-born parents, as these individuals have more family members whose data is available in the Swedish registries. Therefore, this series of analyses included only Swedish-born individuals with 2 Swedish-born parents. We ran the same models as above, excluding region of birth. These models are referred to as Model S1A, S1B, S2, S3, and S4, respectively.

We ran an additional model, Model S5, including an interaction between  $FGRS_{AUD}$  and education and between  $FGRS_{AUD}$  and income, to test whether genetic liability to AUD moderated the associations between education/income and AMC. To be able to interpret the results of the interactions from these multiplicative models on an additive scale, which is often of greater relevance in a public health context [22], we assessed additivity by estimating the relative excess risk due to interaction (RERI) and synergy index (S) [23].

This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guidelines (S1 STROBE Checklist).

#### Results

#### **Descriptive statistics**

Tables 1 and 2 provide descriptions of the female and male cohorts, respectively, in the sample of N = 2,359,338 individuals. In the female cohort, N = 4,253 individuals had an AMC registration, corresponding to 0.37% of the sample. Among males, N = 11,183 (0.93%) had an AMC registration. For both sexes, the distribution of sociodemographic variables, including the primary predictors of interest, varied significantly across the groups without and with AMC. For example, individuals with AMC were less likely to be married than their counterparts without AMC (females: 42.04% versus 56.42%; males: 32.59% versus 50.75%). In addition, among individuals residing in Sweden who were born elsewhere, AMC registrations were disproportionately common only among those from Finland: 2.81% of females and 2.06% of males in the cohort were Finnish, but Finns represented 6.51% of females and 5.54% of males with AMC. In contrast, immigrants from Asia, Africa, and the Middle East had disproportionately few AMC registrations. Incidences of AMC for each variable, per 10,000 person years, are provided in the S1 Table.

The polychoric correlation between income quartile and education level for females was 0.185 (SE-0.001) and for males was 0.198 (SE = 0.001), suggesting that while these

	Total female cohort (within column %/SD) <sup>1</sup>	Without AMC (within column %/SD) <sup>1</sup>	With AMC (within column %/SD) <sup>1</sup>	With AMC (within row %/SD) <sup>2</sup>	Difference test <sup>3</sup> <i>p</i> -value across those without versus with AMC
All	1,162,679	1,158,426	4,253	4,253 (0.37%)	n/a
Unmarried	348,653 (29.99%)	347,278 (29.98%)	1,375 (32.33%)	1,375 (0.39%)	<0.00001
Married	655,346 (56.37%)	653,558 (56.42%)	1,788 (42.04%)	1788 (0.27%)	_
Divorced	152,257 (13.1%)	151,208 (13.05%)	1,049 (24.66%)	1,049 (0.69%)	
Widowed	6,423 (0.55%)	6,382 (0.55%)	41 (0.96%)	41 (0.64%)	
Low education	126,378 (10.87%)	125,395 (10.82%)	983 (23.11%)	983 (0.78%)	<0.00001
Mid education	551,231 (47.41%)	549,021 (47.39%)	2,210 (51.96%)	2,210 (0.4%)	
High education	485,070 (41.72%)	484,010 (41.78%)	1,060 (24.92%)	1,060 (0.22%)	
Low income	150,952 (12.98%)	149,596 (12.91%)	1,356 (31.88%)	1,356 (0.9%)	<0.00001
Low-mid income	275,066 (23.66%)	273,897 (23.64%)	1,169 (27.49%)	1,169 (0.42%)	
High-mid income	352,095 (30.28%)	351,197 (30.32%)	898 (21.11%)	898 (0.26%)	
High income	384,566 (33.08%)	383,736 (33.13%)	830 (19.52%)	830 (0.22%)	
Africa	6,372 (0.55%)	6,363 (0.55%)	9 (0.21%)	9 (0.14%)	<0.00001
Asia	21,140 (1.82%)	21,119 (1.82%)	21 (0.49%)	21 (0.1%)	
Eastern Europe	43,841 (3.77%)	43,707 (3.77%)	134 (3.15%)	134 (0.31%)	
Finland	32,683 (2.81%)	32,406 (2.8%)	277 (6.51%)	277 (0.85%)	
Latin America/ Caribbean	9,299 (0.8%)	9,279 (0.8%)	20 (0.47%)	20 (0.22%)	
Middle East/North Africa	24,546 (2.11%)	24,531 (2.12%)	15 (0.35%)	15 (0.06%)	
Sweden	1,003,382 (86.3%)	999,681 (86.3%)	3,701 (87.02%)	3,701 (0.37%)	
Western countries	21,416 (1.84%)	21,340 (1.84%)	76 (1.79%)	76 (0.35%)	
Internalizing registrations	286,160 (24.61%)	284,452 (24.56%)	1,708 (40.16%)	1,708 (0.6%)	<0.00001
Externalizing registrations	101,851 (8.76%)	100,384 (8.67%)	1,467 (34.49%)	1,467 (1.44%)	<0.00001
AUD registrations	40,717 (3.5%)	38,169 (3.29%)	2,548 (59.91%)	2,548 (6.26%)	<0.00001
Mean follow-up time (years)	18.36 (6.10)	18.36 (6.10)	13.61 (6.51)		<0.00001
Range follow-up time (years)	0.1, 29.0	0.1, 29.0	0.1, 28.9		
Mean birth year	1960.12 (6.04)	1960.13 (6.04)	1957.09 (5.32)		<0.00001
Range birth year	1950, 1970	1950, 1970	1950, 1970		

#### Table 1. Descriptive information on the female cohort, born 1950–1970, with observation beginning at age 40.

<sup>1</sup>For reported percentages, the numerator is the *N* listed in the cell, and the denominator is the *N* in the top row of the same column. For example, there were N = 348,653 unmarried females, which corresponds to 29.99% of the sample (348,653/1,162,679).

<sup>2</sup>For reported percentages, the numerator is the *N* listed in the cell, and the denominator is the *N* in the first column of the same row. For example, there were N = 1,375 unmarried females with AMC, which corresponds to 0.39% of the unmarried females in the sample (1,375/348,653).

<sup>3</sup>For categorical variables, this is a Chi-square; for continuous variables, this is a t test.

AMC, alcohol-related medical conditions; AUD, alcohol use disorder; SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1004359.t001

socioeconomic measures are related, they also provide nonoverlapping information. The <u>S2</u> and <u>S3</u> Tables provide cross-tabulations for females and males, respectively, of income quartile and education level.

# **Crude models**

In crude models unadjusted for any covariates, females with lower educational attainment were at higher risk of AMC (low versus high: HR = 4.48, 95% CIs = 3.66, 5.47, Chi-square

	Total male cohort (col %/SD) <sup>1</sup>	Without AMC (col %/SD) <sup>1</sup>	With AMC (col %/SD) <sup>1</sup>	With AMC (row %/SD) <sup>2</sup>	Difference test <sup>3</sup>
All	1,196,659	1,185,476	11,183	11,184 (0.93%)	n/a
Unmarried	475,629 (39.75%)	470,065 (39.65%)	5,564 (49.75%)	5,564 (1.17%)	< 0.00001
Married	605,286 (50.58%)	601,641 (50.75%)	3,645 (32.59%)	3,645 (0.6%)	
Divorced	113,865 (9.52%)	111,920 (9.44%)	1,945 (17.39%)	1,945 (1.71%)	
Widowed	1,879 (0.16%)	1,850 (0.16%)	29 (0.26%)	29 (1.54%)	
Low education	188,732 (15.77%)	185,959 (15.69%)	2,773 (24.80%)	2,773 (1.47%)	< 0.00001
Mid education	580,381 (48.50%)	574,971 (48.50%)	5,410 (48.38%)	5,410 (0.93%)	
High education	427,546 (35.73%)	424,546 (35.81%)	3,000 (26.83%)	3,000 (0.7%)	
Low income	199,405 (16.66%)	194,905 (16.44%)	4,500 (40.24%)	4,500 (2.26%)	< 0.00001
Low-mid income	284,353 (23.76%)	281,694 (23.76%)	2,659 (23.78%)	2,659 (0.94%)	
High-mid income	375,414 (31.37%)	373,129 (31.47%)	2,285 (20.43%)	2,285 (0.61%)	
High income	337,487 (28.2%)	335,748 (28.32%)	1,739 (15.55%)	1,739 (0.52%)	
Africa	7,735 (0.65%)	7,698 (0.65%)	37 (0.33%)	37 (0.48%)	< 0.00001
Asia	15,810 (1.32%)	15,739 (1.33%)	71 (0.63%)	71 (0.45%)	
Eastern Europe	33,009 (2.76%)	32,759 (2.76%)	250 (2.24%)	250 (0.76%)	
Finland	24,624 (2.06%)	24,004 (2.02%)	620 (5.54%)	620 (2.52%)	
Latin America/Caribbean	8,718 (0.73%)	8,664 (0.73%)	54 (0.48%)	54 (0.62%)	
Middle East/North Africa	35,680 (2.98%)	35,552 (3%)	128 (1.14%)	128 (0.36%)	
Sweden	1,047,228 (87.51%)	1,037,379 (87.51%)	9,849 (88.07%)	9,849 (0.94%)	
Western countries	23,855 (1.99%)	23,681 (2%)	174 (1.56%)	174 (0.73%)	
Internalizing registrations	177,998 (14.87%)	174,804 (14.75%)	3,194 (28.57%)	3,194 (1.79%)	< 0.00001
Externalizing registrations	276,156 (23.08%)	269,886 (22.77%)	6,270 (56.07%)	6,270 (2.27%)	< 0.00001
AUD registrations	109,221 (9.13%)	101,499 (8.56%)	7,722 (69.05%)	7,722 (7.07%)	< 0.00001
Mean follow-up time (years)	18.17 (6.03)	18.01 (6.11)	13.47 (6.55)		< 0.00001
Range follow-up time (years)	0.1, 29.0	0.1, 29.0	0.1, 28.9		
Mean birth year	1960.17 (6.03)	1860.20 (6.03)	1957.02 (5.26)		<0.00001
Range birth year	1950, 1970	1950, 1970	1950, 1970		

#### Table 2. Descriptive information on the male cohort, born 1950-1970, with observation beginning at age 40.

<sup>1</sup>For reported percentages, the numerator is the *N* listed in the cell, and the denominator is the *N* in the top row of the same column. For example, there were N = 475,632 unmarried males, which corresponds to 39.75% of the sample (475,632/1,196,663).

<sup>2</sup>For reported percentages, the numerator is the *N* listed in the cell, and the denominator is the *N* in the first column of the same row. For example, there were N = 5,564 unmarried males with AMC, which corresponds to 1.17% of the unmarried males in the sample (5,564/475,632).

<sup>3</sup>For categorical variables, this is a Chi-square; for continuous variables, this is a *t* test.

AMC, alcohol-related medical conditions; AUD, alcohol use disorder; SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1004359.t002

p < 0.001; mid versus high: HR = 1.98, 95% CIs = 1.66, 2.34, p < 0.001); the same was true for males, at lower effect sizes (low versus high: HR = 2.27, 95% CIs = 2.01, 2.55, p < 0.001; mid versus high: HR = 1.46, 95% CIs = 1.32, 2.61, p < 0.001). Likewise, females in lower income quartiles had higher risk of AMC (income quartile 1 versus 4: HR = 11.52 [9.30, 14.27], p < 0.001; income quartile 2 versus 4: HR = 3.90 [3.12, 4.87], p < 0.001; income quartile 3 versus 4: HR = 1.60 [1.26, 2.03], p = 0.001). Results for males were comparable (income quartile 1 versus 4: HR = 11.95 [10.41, 13.72], p < 0.001; income quartile 2 versus 4: HR = 1.63 [1.39, 1.91], p < 0.001).

#### Multivariable models

We next estimated the association between educational attainment (in Model 1A) or income at age 40 (in Model 1B) with risk of AMC, adjusting only for sociodemographic covariates. We



#### Time dependent HR

**Fig 1. HRs between education level and/or income with risk of AMCs, across time, from Models 1A/1B–Model 4**. Models 1A and 1B include education level or income, respectively, as the primary predictors of interest, and are adjusted for birth year, marital status, and region of origin. Model 2 includes both education and income level with the aforementioned covariates. Model 3 further adjusts for internalizing and externalizing registrations. Model 4 further adjusts for AUD registrations. Results for females are presented in the top panel and for males in the bottom panel. The horizontal dashed gray line at an HR of 1 represents the null hypothesis. AMC, alcohol-related medical condition; AUD, alcohol use disorder; HR, hazard ratio.

https://doi.org/10.1371/journal.pmed.1004359.g001

provide snapshots of the HRs for education and income at 4 time points: time 0 (age 40), after 5 years of observation (age 45), after 10 years of observation (age 50), and after 15 years (age 55 or older). As shown in the **Fig 1**, in which estimates from Models 1A and 1B are superimposed for ease of illustration, individuals with lower levels of education or income at age 40 were at substantially increased risk for AMC. Complete results, including for all covariates, are provided in S4 and S5 Tables.

Among females (Fig 1 and S4 Table), having the lowest level of education was associated with a nearly 5-fold increase in risk of AMC in Model 1A (HR = 4.67; 95% CI 3.82, 5.72; p < 0.001). Relative to individuals with the highest level of education, those with mid- or low-level education at age 40 were still at increased risk of AMC 15 years later (HRs = 1.68 [95% CI s 1.56, 1.81], p < 0.001; and 2.84 [95% CI s 2.60, 3.11), p < 0.001, respectively). HRs for the 3 lower quartiles of income were also elevated (Model 1B; Fig 1 and S5 Table). The magnitude of effect declined across observation time, most notably for the lowest quartile (from HR = 11.26 at observation onset to HR = 3.86, 15 years later), and was more pronounced for income than for education. We observed a general dose-response pattern for levels of education/income and for time elapsed since the start of observation; i.e., the strongest effects were observed at the start of observation and income levels.

HRs were lower, but still consistently >1, for males (Fig 1 and S4 Table). In males, being in the lowest education category (Model 1A) was associated with a 2-fold increase in risk of AMC (HR = 2.04, 95% CIs 1.81, 2.30; p < 0.001) at the start of observation. Over time, these

estimates declined, but remained above 1 (HRs = 1.26 to 1.57 for those with mid-level and low-level education). Similar to females, the effect size for being in the lowest income quartile was high: HR = 10.21 (95% CIs 8.86, 11.77; p < 0.001; Model 1B; Fig 1 and S5 Table). Individuals in the 2 middle quartiles were at lower, but still elevated, risk of AMC. Also consistent with females, the HRs for education were more stable over time than those for income.

In Model 2 (S6 Table), education and income were jointly included as predictors. The HRs were attenuated, but only slightly, indicating that these measures of socioeconomic position account for independent components of AMC risk. We next adjusted for psychiatric illness using time-varying covariates, in Model 3 (S7 Table). Among females, HRs for ID and ED were associated with an approximately 2.6- to 3.7-fold increase in AMC risk. We observed a slight attenuation in the effect sizes of education and income from Model 2. The lowest income quartile remained at high risk of AMC at the start of observation (HR = 7.80, 95% CIs 6.24, 9.75; p < 0.001), declining to HR = 2.88, 15 years later. A similar pattern was observed for males, among whom HRs were approximately 2.1 to 3.1 for ID and ED. For males, having a mid- or low-level education was associated with a 1.21- to 1.63-fold increase in AMC risk. Being in the lowest income quartile was still strongly associated with AMC (HR = 7.76, 95% CIs 6.73, 8.95; p < 0.001 at start of observation), though the effects were attenuated as time progressed and being in the third quartile 15 years after the onset of observation was not significantly associated with higher AMC risk.

Finally, in Model 4, we adjusted for AUD as a time-varying covariate. AUD was prominently associated with AMC risk: the associated HRs were HR = 34.00 (95% CIs 31.73, 36.42; p < 0.001) for females and HR = 10.71 (95% CIs 10.28, 11.15; p < 0.001) for males (S8 Table). For both sexes, the inclusion of AUD led to lower HRs for education and income relative to previous models. However, with the exception of the estimate for individuals in the third income quartile, 15 years after the start of observation, all estimates remained significantly >1. We continued to observe a dose-response pattern for income quartiles and as a function of time elapsed since onset of observation in most cases, the exception being that the effect of having mid-level education at age 40 was relatively stable across the observation period. This pattern was clearer among males.

#### Secondary analysis

We pursued a secondary set of analyses that corresponded to the original models, replacing region of origin with FGRS<sub>AUD</sub>, as described in the Methods. In Models S1A/S1B through Model S4, the effect of  $FGRS_{AUD}$  ranged from HR = 1.17 to 1.39 for females and HR = 1.21to 1.36 for males (<u>S9–S13</u> Tables) per standard deviation increase. The HRs for education level and income did not differ substantially from the corresponding HRs from the original Models 1 to 4. We added a model with an interaction term between education/income and FGRS<sub>AUD</sub>, Model S5 to evaluate the possibility that level of aggregate genetic liability to AUD leads to differential susceptibility to the risky effects of lower education or income. The main effects of all predictors and covariates are reported in S14 Table, and we report the RERI and S (estimates of interactions on the additive scale) in S15 Table. For females, there was no evidence that FGRS<sub>AUD</sub> moderated the association between education level and AMC risk. However, RERI and S estimates were consistent with an interaction between FGRS<sub>AUD</sub> and the first and second income quartiles, such that individuals at higher genetic liability to AUD were more susceptible to the adverse impact of lower income on risk for AMC. Similarly, in males, we observed no interaction between FGRS<sub>AUD</sub> and education level, but the association between income and AMC was significantly moderated by FGRS<sub>AUD</sub>.

## Discussion

In the current study, we used nationwide longitudinal Swedish registry data to evaluate a research question of clinical and social importance: whether 2 socioeconomic indicators, education and income, are related to risk of alcohol-related medical conditions. Even after accounting for other sociodemographic measures, psychopathology, and the role of AUD itself, individuals with lower levels of education and income had an increased risk of AMC. For education, the associations were slightly higher among females. For both sexes, the magnitude of effect was higher for lower income level than for lower education, and generally decreased across the observation period but in nearly every case remained elevated even 15 years after the start of observation. Furthermore, the adverse impact of low income was exacerbated among females and males with higher genetic liability to AUD. These findings demonstrate that lower education and income, though related, have at least partially independent and persistent associations with risk of AMC, which could contribute to exacerbated health disparities related to SEP. This effect is only slightly mediated by AUD, suggesting that individuals of lower SEP may warrant particular clinical attention, even within the already high-risk group of individuals with AUD.

The magnitude of effect for income was considerably higher at the onset of observation (age 40) than the corresponding effect of education, and its attenuation was more pronounced across the observation period. The change over time is likely due at least in part to shifts in income during middle to later adulthood: For example, if income increases after age 40, the effect of one's prior income on subsequent health may be superseded by that of their more recent income. However, income at the point of AMC registration might not be the most relevant measure, as AMC is typically the result of chronically high alcohol consumption that began years before.

As noted above, previous research has found that drinking patterns vary by SEP, with those of lower SEP more likely to engage in heavy consumption [11–14,16], which is strongly linked to risk of AMC [24–26]. Information on consumption levels is not available in the Swedish nationwide registries, precluding our ability to test whether the associations between education/income remain when accounting for chronic heavy drinking, for example. However, our finding that these associations persist when controlling for AUD—itself closely associated with consumption levels [27,28]—is consistent with prior evidence that individuals of lower SEP are at disproportionately high risk of AMC even after accounting for heavy drinking [16].

With respect to the discrepancy between effect sizes for education and income, some prior research suggests that education is a more potent risk factor for the onset of chronic health conditions, while income more strongly predicts their progression [29]. A study that included the current cohort reported more protective effects between education and AUD than between income and AUD [30], though differences were not as pronounced as observed in the current study. Although these nuances remain to be investigated in future research, the current study clearly indicates that education and income are independently associated with AMC risk, though in assessing future risk, clinicians should pay particular attention to income.

What might contribute to this unequal risk across socioeconomic strata? Some studies indicate that individuals of lower SEP are less likely to engage in preventive health care behaviors [31]. Even in countries with socialized medical care, where access should be less of a barrier, these discrepancies may be due to health literacy [32] and less expendable time [33]. A study of large Finnish and British cohorts found that the association between lower SEP and poor health outcomes was driven in part by psychiatric and substance use disorders, which in turn led to downstream medical problems [34]. Still, accounting for other AUD registrations in the current study resulted in only slight attenuations to the observed associations, providing evidence that SEP indicators do not merely exert their effects through AUD. In addition to the effects of our primary predictors of interest, we note 3 incidental findings that merit further investigation in the context of models specifically designed to replicate our observed associations. First, in fully adjusted models (Model 4 and Model S4), unmarried females were at reduced risk of AMC relative to married females, while among males, married individuals were consistently at lowest AMC risk. Prior studies have supported a protective effect of marriage for alcohol use, AUD, and other psychiatric outcomes [35–39], suggesting that AMC is a relative outlier, and only for females. This could be due to a stronger negative impact of husbands' alcohol use on wives' alcohol-related outcomes [40], an exposure that would not apply to unmarried females. The dynamics at play are likely to be challenging to dissect using registry data, but merit follow-up.

Second, with the exception of those from Finland, immigrants were at lower risk of AMC than their Swedish-born counterparts. One potential explanation is that immigrants from many regions have lower levels of alcohol consumption than Swedes and many other European nations [41]. A study of immigrant cohorts in Sweden showed that foreign-born Swedes from Asia and the Middle East (males and females) and from African and Eastern Europe (females) had lower rates of AUD than the general population [42]. Finns, who constitute one of the largest immigrant groups in Sweden, had the highest AUD rates (males and females). Finland is included in the so-called "vodka belt," i.e., those Northern countries where the use of stronger alcoholic beverages (spirits) and binge drinking are more common than in other countries [43]. Prior studies also document Finns' excess mortality in Sweden across multiple medical conditions [44] and their self-reported poor health compared to other Swedes and to Finns living in Finland [45], which together suggests their greater risk of AMC signals wider health inequities. Additionally, immigrant groups may be less likely to use medical care, even in nations where it is broadly accessible. For example, foreign-born patients are less likely than Swedish-born patients to pick up their medication for treating AUD from the pharmacy [46]. It then is possible that registry data could potentially underrepresent the extent of AMC among immigrants. If this is the case, additional outreach may be necessary to ensure that immigrants are adequately screened for risky drinking and treated for AUD to prevent the development of AUD and AMC.

Third, we found that individuals with higher genetic liability to AUD were more susceptible to the adverse impact of low income, though this was not the case for those with lower levels of educational attainment. This could reflect the stronger overall effect size of income level, and potentially due to the fact that the estimate is based on year 0, which is when the effect of income was most pronounced; replication in other samples, and perhaps using complementary methods (e.g., molecular polygenic scores) is needed. However, these results do suggest that clinicians should be attentive to the potentially combinatorial effects of low-income and dense family history of AUD: Patients subject to both conditions may warrant additional screening to ensure that their alcohol use does not result in adverse health outcomes.

Strengths of the current analysis include the use of nationwide, longitudinal databases with minimal bias; comparison of 2 related measures of SEP with unique contributions to AMC risk; and our ability to control for important covariates such as comorbid psychopathology and genetic liability to AUD. Our findings must still be viewed in the context of several limitations. First, income can change frequently, but to facilitate modeling we focused on income at the beginning of observation (age 40). Including repeated measures of income could have resulted in different findings, e.g., the effect size of income might decline less rapidly than in the present study. Second, while individuals with an AMC registration are likely to meet criteria for AUD, approximately 30% to 40% of those with AMC registrations did not have another registration for AUD (e.g., through the ICD-10 F10 codes), suggesting inconsistency in how clinicians record patients' AUD. Third, due to the nature of FGRS<sub>AUD</sub>, these scores are less

precise among immigrants, precluding our ability to confidently estimate the extent to which aggregate genetic liability to AUD contributes directly to AMC and/or mediates/moderates the effects of income or education among those born outside of Sweden. Fourth, we were unable to directly account for differences in alcohol consumption, which is correlated with AUD, and which may differ by SEP. The medical community's perspective on the impact of alcohol consumption at different levels is variable and subject to revision. For example, while many studies support a J-shaped association between consumption and certain health outcomes [47], other considerations (e.g., the climate-based impacts of the alcohol industry) have led some entities to recommend abstinence [48]. These issues are beyond the scope of the current study but warrant further consideration for public health considerations.

Fifth, we cannot firmly ascribe the observed associations to a causal path: Education and income are correlated with an array of psychosocial factors that could act as confounders, and complementary methodological approaches are necessary to provide additional insight. Finally, alcohol consumption varies across countries and cultures, and the current findings might not generalize to other contexts. For example, although the drinking age is as low as 18 in Sweden, alcohol access is constrained by relatively high costs and by the Systembolaget, which controls liquor sales and has limited hours of operation. Greater alcohol accessibility in other countries, especially where costs are lower, could exacerbate the risks of AMC associated with lower SEP. Furthermore, the sex differences observed in the current study might not be replicated in societies with lower sexual egalitarianism. Additional research in other cultural contexts is therefore essential.

In summary, we provide evidence that individuals with lower levels of education and/or income are more likely to suffer from AMC, even after accounting for differences in AUD, comorbid psychopathology, and aggregate genetic liability. Income generally plays a more prominent role, though the effects of both SEP indicators persist across time. Secondary observations suggest that more research is needed to better understand differences related to sex, marital status, and region of origin. These findings contribute to a growing body of literature on health disparities as a function of socioeconomic resources and suggest that individuals with fewer such resources could benefit from additional clinical attention regarding the risks associated with problematic alcohol use.

# Supporting information

**S1** Methods. Description of registry resources and analytic methods. (DOCX)

**S1** Table. Incidence rates of alcohol-related medical conditions, reported as number of new cases per 10,000 person years, for variables that are constant over time. (DOCX)

**S2 Table.** Cross-tabulation of income quartile and education level for females in the full sample (top panel) and among those with an alcohol-related medical condition (AMC; bottom panel).

(DOCX)

**S3 Table.** Cross-tabulation of income quartile and education level for males in the full sample (top panel) and among those with an alcohol-related medical condition (AMC; bottom panel). (DOCX)

S4 Table. Complete results for Model 1A for females and males, testing the association between education level and alcohol-related medical conditions. Hazard ratios, 95%

confidence intervals, and *p*-values from Chi-square tests are presented. The primary predictor of interest (here, education level) was modeled using a time-varying coefficient, with a linear term for time. Below, we provide snapshots of hazard ratios for education at 4 time points: at the beginning of observation (time 0), after 5 years, after 10 years, and after 15 years. (DOCX)

S5 Table. Complete results for Model 1B for females and males, testing the association between income and alcohol-related medical conditions. Hazard ratios, 95% confidence intervals, and *p*-values from Chi-square tests are presented. The primary predictor of interest (here, income) was modeled using a time-varying coefficient, with a linear term for time. Below, we provide snapshots of hazard ratios for income at 4 time points: at the beginning of observation (time 0), after 5 years, after 10 years, and after 15 years. (DOCX)

S6 Table. Complete results for Model 2 for females and males, testing the associations between education level and income with alcohol-related medical conditions. Hazard ratios, 95% confidence intervals, and p-values from Chi-square tests are presented. The primary predictors of interest (education level and income) were modeled using time-varying coefficients, with a linear term for time. Below, we provide snapshots of hazard ratios for education and income at 4 time points: at the beginning of observation (time 0), after 5 years, after 10 years, and after 15 years.



S7 Table. Complete results for Model 3 for females and males, testing the associations between education level and income with alcohol-related medical conditions. Hazard ratios, 95% confidence intervals, and Chi-square p-values are presented. The primary predictors of interest (education level and income) were modeled using time-varying coefficients, with a linear term for time. Below, we provide snapshots of hazard ratios for education and income at 4 time points: at the beginning of observation (time 0), after 5 years, after 10 years, and after 15 years.

(DOCX)

S8 Table. Complete results for Model 4 for females and males, testing the associations between education level and income with alcohol-related medical conditions. Hazard ratios, 95% confidence intervals, and Chi-square p-values are presented. The primary predictors of interest (education level and income) were modeled using time-varying coefficients, with a linear term for time. Below, we provide snapshots of hazard ratios for education and income at 4 time points: at the beginning of observation (time 0), after 5 years, after 10 years, and after 15 years.

(DOCX)

S9 Table. Complete results for Model S1A for females and males, testing the association between education level and alcohol-related medical conditions. Hazard ratios, 95% confidence intervals, and Chi-square *p*-values are presented. The primary predictor of interest (here, education level) was modeled using a time-varying coefficient, with a linear term for time. Below, we provide snapshots of hazard ratios for education at 4 time points: at the beginning of observation (time 0), after 5 years, after 10 years, and after 15 years. These secondary analyses were limited to the subsample born in Sweden with 2 Swedish-born parents to improve the precision of the family genetic risk score for alcohol use disorder; accordingly, region of interest is excluded as a covariate. (DOCX)

**S10 Table. Complete results for Model S1B for females and males, testing the association between income and alcohol-related medical conditions.** Hazard ratios and 95% confidence intervals are presented. The primary predictor of interest (here, income) was modeled using a time-varying coefficient, with a linear term for time. Below, we provide snapshots of hazard ratios for income at 4 time points: at the beginning of observation (time 0), after 5 years, after 10 years, and after 15 years. These secondary analyses were limited to the subsample born in Sweden with 2 Swedish-born parents to improve the precision of the family genetic risk score for alcohol use disorder; accordingly, region of interest is excluded as a covariate. (DOCX)

**S11 Table.** Complete results for Model S2 for females and males, testing the associations between education level and income with alcohol-related medical conditions. Hazard ratios, 95% confidence intervals, and Chi-square *p*-values are presented. The primary predictors of interest (education level and income) were modeled using time-varying coefficients, with a linear term for time. Below, we provide snapshots of hazard ratios for education level and income at 4 time points: at the beginning of observation (time 0), after 5 years, after 10 years, and after 15 years. These secondary analyses were limited to the subsample born in Sweden with 2 Swedish-born parents to improve the precision of the family genetic risk score for alcohol use disorder; accordingly, region of interest is excluded as a covariate. (DOCX)

**S12 Table.** Complete results for Model S3 for females and males, testing the associations between education level and income with alcohol-related medical conditions. Hazard ratios, 95% confidence intervals, and Chi-square *p*-values are presented. The primary predictors of interest (education level and income) were modeled using time-varying coefficients, with a linear term for time. Below, we provide snapshots of hazard ratios for education level and income at 4 time points: at the beginning of observation (time 0), after 5 years, after 10 years, and after 15 years. These secondary analyses were limited to the subsample born in Sweden with 2 Swedish-born parents to improve the precision of the family genetic risk score for alcohol use disorder; accordingly, region of interest is excluded as a covariate. (DOCX)

**S13 Table.** Complete results for Model S4 for females and males, testing the associations between education level and income with alcohol-related medical conditions. Hazard ratios, 95% confidence intervals, and Cho-square *p*-values are presented. The primary predictors of interest (education level and income) were modeled using time-varying coefficients, with a linear term for time. Below, we provide snapshots of hazard ratios for education level and income at 4 time points: at the beginning of observation (time 0), after 5 years, after 10 years, and after 15 years. These secondary analyses were limited to the subsample born in Sweden with 2 Swedish-born parents to improve the precision of the family genetic risk score for alcohol use disorder; accordingly, region of interest is excluded as a covariate. (DOCX)

**S14 Table.** Complete results for Model S5 for females and males, testing the associations between education level and income with alcohol-related medical conditions. Hazard ratios, 95% confidence intervals, and Chi-square *p*-values are presented. The primary predictors of interest (education level and income) were modeled using time-varying coefficients, with a linear term for time. Below, we provide snapshots of hazard ratios for education level and income at 4 time points: at the beginning of observation (time 0), after 5 years, after 10 years, and after 15 years. These secondary analyses were limited to the subsample born in Sweden with 2 Swedish-born parents to improve the precision of the family genetic risk score for

alcohol use disorder; accordingly, region of interest is excluded as a covariate. This model includes an interaction term between  $FGRS_{AUD}$  and each level of education and income; those results are in <u>S15 Table</u>.

(DOCX)

S15 Table. Model S5 includes an interaction term between  $FGRS_{AUD}$  and each level of education and income (main effects are provided in S11 Table). We present these as interactions on the additive scale through the use of the relative excess risk due to interaction (RERI) and synergy index (S), along with corresponding 95% confidence intervals; the *p*-value is based on a Chi-square test of the interaction term. These terms are estimated based on the effect of education and income at time 0.

(DOCX)

**S1 STROBE Checklist. STROBE checklist.** (DOCX)

# **Author Contributions**

Conceptualization: Alexis C. Edwards, Sara Larsson Lönn.

Data curation: Sara Larsson Lönn.

Formal analysis: Sara Larsson Lönn.

Funding acquisition: Jan Sundquist, Kenneth S. Kendler, Kristina Sundquist.

Methodology: Alexis C. Edwards, Sara Larsson Lönn.

**Project administration:** Alexis C. Edwards, Jan Sundquist, Kenneth S. Kendler, Kristina Sundquist.

Writing - original draft: Alexis C. Edwards, Sara Larsson Lönn.

Writing – review & editing: Alexis C. Edwards, Sara Larsson Lönn, Karen G. Chartier, Séverine Lannoy, Jan Sundquist, Kenneth S. Kendler, Kristina Sundquist.

#### References

- 1. World Health Organization. Alcohol: World Health Organization. 2022 [Accessed 08/23/2023]. Available from: https://www.who.int/news-room/fact-sheets/detail/alcohol.
- Stahre M, Roeber J, Kanny D, Brewer RD, Zhang X. Contribution of excessive alcohol consumption to deaths and years of potential life lost in the United States. Prev Chronic Dis. 2014; 11:E109. Epub 20140626. https://doi.org/10.5888/pcd11.130293 PMID: 24967831; PubMed Central PMCID: PMC4075492.
- World Health Organization. Alcohol country fact sheet. Sweden: World Health Organization; 2019 [Accessed 08/23/2023]. Available from: <u>https://cdn.who.int/media/docs/librariesprovider2/country-sites/</u> sweden/achp\_fs\_sweden.pdf?sfvrsn=f12c768e\_3&download=true.
- Thomas NS, Kuo SI, Aliev F, McCutcheon VV, Meyers JM, Chan G, et al. Alcohol use disorder, psychiatric comorbidities, marriage and divorce in a high-risk sample. Psychol Addict Behav. 2022; 36 (4):364–74. Epub 20220526. https://doi.org/10.1037/adb0000840 PMID: 35617219; PubMed Central PMCID: PMC9247836.
- Whiting D, Lichtenstein P, Fazel S. Violence and mental disorders: a structured review of associations by individual diagnoses, risk factors, and risk assessment. Lancet Psychiatry. 2021; 8(2):150–61. Epub 20201020. https://doi.org/10.1016/S2215-0366(20)30262-5 PMID: 33096045.
- Li J, Wang H, Li M, Shen Q, Li X, Zhang Y, et al. Effect of alcohol use disorders and alcohol intake on the risk of subsequent depressive symptoms: a systematic review and meta-analysis of cohort studies. Addiction. 2020; 115(7):1224–43. https://doi.org/10.1111/add.14935 PMID: 31837230

- Edwards AC, Ohlsson H, Moscicki E, Crump C, Sundquist J, Kendler KS, et al. Alcohol use disorder and non-fatal suicide attempt: findings from a Swedish National Cohort Study. Addiction. 2022; 117 (1):96–105. Epub 20210712. https://doi.org/10.1111/add.15621 PMID: 34159695; PubMed Central PMCID: PMC Journal–In Process.
- Edwards AC, Ohlsson H, Sundquist J, Sundquist K, Kendler KS. Alcohol Use Disorder and Risk of Suicide in a Swedish Population-Based Cohort. Am J Psychiatry. 2020; 177(7):627–34. Epub 20200312. https://doi.org/10.1176/appi.ajp.2019.19070673 PMID: 32160767; PubMed Central PMCID: PMC8887810.
- Sacks JJ, Gonzales KR, Bouchery EE, Tomedi LE, Brewer RD. 2010 National and State Costs of Excessive Alcohol Consumption. Am J Prev Med. 2015; 49(5):e73–e9. Epub 2015/10/20. https://doi. org/10.1016/j.amepre.2015.05.031 PMID: 26477807.
- Collins SE. Associations Between Socioeconomic Factors and Alcohol Outcomes. Alcohol Res. 2016; 38(1):83–94. Epub 2016/05/10. PMID: 27159815; PubMed Central PMCID: PMC4872618.
- van Oers JA, Bongers IM, van de Goor LA, Garretsen HF. Alcohol consumption, alcohol-related problems, problem drinking, and socioeconomic status. Alcohol Alcohol. 1999; 34 (1):78–88. <u>https://doi.org/ 10.1093/alcalc/34.1.78</u> PMID: 10075406.
- Huckle T, You RQ, Casswell S. Socio-economic status predicts drinking patterns but not alcohol-related consequences independently. Addiction. 2010; 105(7):1192–202. Epub 20100427. <u>https://doi.org/10.1111/j.1360-0443.2010.02931.x</u> PMID: 20456295.
- Knupfer G. The prevalence in various social groups of eight different drinking patterns, from abstaining to frequent drunkenness: analysis of 10 U.S. surveys combined. Br J Addict. 1989; 84(11):1305–1318. https://doi.org/10.1111/j.1360-0443.1989.tb00732.x PMID: 2597807.
- Kendler KS, Gardner CO, Hickman M, Heron J, Macleod J, Lewis G, et al. Socioeconomic status and alcohol-related behaviors in mid- to late adolescence in the Avon Longitudinal Study of Parents and Children. J Stud Alcohol Drugs. 2014; 75(4):541–545. https://doi.org/10.15288/jsad.2014.75.541 PMID: 24988252; PubMed Central PMCID: 4108596.
- Probst C, Roerecke M, Behrendt S, Rehm J. Socioeconomic differences in alcohol-attributable mortality compared with all-cause mortality: a systematic review and meta-analysis. Int J Epidemiol. 2014; 43 (4):1314–27. Epub 20140311. <u>https://doi.org/10.1093/ije/dyu043</u> PMID: <u>24618188</u>; PubMed Central PMCID: PMC4258771.
- Thern E, Landberg J. Understanding the differential effect of alcohol consumption on the relation between socio-economic position and alcohol-related health problems: results from the Stockholm Public Health Cohort. Addiction. 2021; 116(4):799–808. Epub 20200831. <u>https://doi.org/10.1111/add.</u> 15213 PMID: 32738005; PubMed Central PMCID: PMC8048434.
- Kendler KS, Ohlsson H, Karriker-Jaffe KJ, Sundquist J, Sundquist K. Social and economic consequences of alcohol use disorder: a longitudinal cohort and co-relative analysis. Psychol Med. 2017; 47 (5):925–935. https://doi.org/10.1017/S0033291716003032 PMID: 27917730; PubMed Central PMCID: PMC5340576.
- Kendler KS, Larsson Lonn S, Morris NA, Sundquist J, Langstrom N, Sundquist K. A Swedish national adoption study of criminality. Psychol Med. 2014; 44(9):1913–25. Epub 2013/11/05. https://doi.org/10. 1017/S0033291713002638 PMID: 24180693; PubMed Central PMCID: PMC4009388.
- Kendler KS, Ohlsson H, Sundquist J, Sundquist K. The patterns of family genetic risk scores for eleven major psychiatric and substance use disorders in a Swedish national sample. Transl Psychiatry. 2021; 11(1):326. Epub 2021/05/29. https://doi.org/10.1038/s41398-021-01454-z PMID: 34045441.
- Krebs MD, Appadurai V, Hellberg K-LG, Ohlsson H, Steinbach J, Petersen E, et al. The relationship between genotype- and phenotype-based estimates of genetic liability to human psychiatric disorders, in practice and in theory. medRxiv. 2023:2023.06.19.23291606. <u>https://doi.org/10.1101/2023.06.19</u>. 23291606
- 21. National Institutes of Health. NIH Policy on Sex as a Biological Variable: Office of Research on Women's Health. 2023 [Accessed 2023/12/07]. Available from: https://orwh.od.nih.gov/sex-gender/orwhmission-area-sex-gender-in-research/nih-policy-on-sex-as-biological-variable.
- Kendler KS, Gardner CO. Interpretation of interactions: guide for the perplexed. Br J Psychiatry. 2010; 197(3):170–1. Epub 2010/09/03. https://doi.org/10.1192/bjp.bp.110.081331 [pii] PMID: 20807958.
- Li R, Chambless L. Test for Additive Interaction in Proportional Hazards Models. Ann Epidemiol. 2007; 17(3):227–36. https://doi.org/10.1016/j.annepidem.2006.10.009 PMID: 17320789
- Roerecke M, Vafaei A, Hasan OSM, Chrystoja BR, Cruz M, Lee R, et al. Alcohol Consumption and Risk of Liver Cirrhosis: A Systematic Review and Meta-Analysis. Am J Gastroenterol. 2019; 114(10):1574– 1586. https://doi.org/10.14309/ajg.0000000000340 PMID: 31464740; PubMed Central PMCID: PMC6776700.

- Simpson RF, Hermon C, Liu B, Green J, Reeves GK, Beral V, et al. Alcohol drinking patterns and liver cirrhosis risk: analysis of the prospective UK Million Women Study. Lancet Public Health. 2019; 4(1): e41–e8. Epub 20181122. https://doi.org/10.1016/S2468-2667(18)30230-5 PMID: 30472032; PubMed Central PMCID: PMC6323353.
- Samokhvalov AV, Rehm J, Roerecke M. Alcohol Consumption as a Risk Factor for Acute and Chronic Pancreatitis: A Systematic Review and a Series of Meta-analyses. eBioMedicine. 2015; 2(12):1996– 2002. https://doi.org/10.1016/j.ebiom.2015.11.023 PMID: 26844279
- Dawson DA, Grant BF, Stinson FS, Zhou Y. Effectiveness of the derived Alcohol Use Disorders Identification Test (AUDIT-C) in screening for alcohol use disorders and risk drinking in the US general population. Alcohol Clin Exp Res. 2005; 29(5):844–854. https://doi.org/10.1097/01.alc.0000164374.32229.a2 PMID: 15897730.
- Sanchez-Roige S, Palmer AA, Fontanillas P, Elson SL, 23andMe Research Team tSUDWGotPGC, Adams MJ, et al. Genome-Wide Association Study Meta-Analysis of the Alcohol Use Disorders Identification Test (AUDIT) in Two Population-Based Cohorts. Am J Psychiatry. 2019; 176(2):107–18. Epub 2018/10/20. https://doi.org/10.1176/appi.ajp.2018.18040369 PMID: 30336701; PubMed Central PMCID: PMC6365681.
- Herd P, Goesling B, House JS. Socioeconomic Position and Health: The Differential Effects of Education versus Income on the Onset versus Progression of Health Problems. J Health Soc Behav. 2007; 48 (3):223–238. https://doi.org/10.1177/002214650704800302 PMID: 17982865.
- Calling S, Ohlsson H, Sundquist J, Sundquist K, Kendler KS. Socioeconomic status and alcohol use disorders across the lifespan: A co-relative control study. PLoS ONE. 2019; 14(10):e0224127. Epub 20191017. https://doi.org/10.1371/journal.pone.0224127 PMID: 31622449; PubMed Central PMCID: PMC6797188.
- Vernekar N, Batchelor P, Heilmann A. Adult self-reported attendance for dental check-ups over a 16year period in the UK. Br Dent J. 2019; 226(11):883–888. https://doi.org/10.1038/s41415-019-0366-8 PMID: 31203343.
- 32. Svendsen MT, Bak CK, Sorensen K, Pelikan J, Riddersholm SJ, Skals RK, et al. Associations of health literacy with socioeconomic position, health risk behavior, and health status: a large national population-based survey among Danish adults. BMC Public Health. 2020; 20(1):565. Epub 20200428. https://doi.org/10.1186/s12889-020-08498-8 PMID: 32345275; PubMed Central PMCID: PMC7187482.
- Bick A, Fuchs-Schündeln N, Lagakos D. How Do Hours Worked Vary with Income? Cross-Country Evidence and Implications. Am Econ Rev. 2018; 108(1):170–199. https://doi.org/10.1257/aer.20151720
- Kivimaki M, Batty GD, Pentti J, Shipley MJ, Sipila PN, Nyberg ST, et al. Association between socioeconomic status and the development of mental and physical health conditions in adulthood: a multi-cohort study. Lancet Public Health. 2020; 5(3):e140–e9. Epub 20200131. https://doi.org/10.1016/S2468-2667 (19)30248-8 PMID: 32007134.
- **35.** Plant M, Miller P, Plant M, Kuntsche S, Gmel G, Ahlström S, et al. Marriage, cohabitation and alcohol consumption in young adults: an international exploration. J Subst Use. 2008; 13(2):83–98.
- Leonard KE, Rothbard JC. Alcohol and the marriage effect. J Stud Alcohol Suppl. 1999; 13:139–146. https://doi.org/10.15288/jsas.1999.s13.139 PMID: 10225498
- Kendler KS, Lonn SL, Salvatore J, Sundquist J, Sundquist K. Effect of Marriage on Risk for Onset of Alcohol Use Disorder: A Longitudinal and Co-Relative Analysis in a Swedish National Sample. Am J Psychiatry. 2016; 173(9):911–918. https://doi.org/10.1176/appi.ajp.2016.15111373 PMID: 27180900; PubMed Central PMCID: PMC5008987.
- Weisenhorn DA, Frey LM, van de Venne J, Cerel J. Suicide Exposure and Posttraumatic Stress Disorder: Is Marriage a Protective Factor for Veterans? J Child Fam Stud. 2016; 26(1):161–167. <u>https://doi.org/10.1007/s10826-016-0538-y</u>
- Jaffe DH, Manor O, Eisenbach Z, Neumark YD. The protective effect of marriage on mortality in a dynamic society. Ann Epidemiol. 2007; 17(7):540–7. Epub 20070416. <u>https://doi.org/10.1016/j.</u> annepidem.2006.12.006 PMID: 17434751.
- Kendler KS, Lönn SL, Salvatore J, Sundquist J, Sundquist K. The Origin of Spousal Resemblance for Alcohol Use Disorder. JAMA Psychiatry. 2018. <u>https://doi.org/10.1001/jamapsychiatry.2017.4457</u> PMID: 29417130
- World Health Organization. Alcohol, total per capita (15+) consumption (in litres of pure alcohol) (SDG. Indicator. 2023;3.5.2). Accessed 2023/09/26.
- Chartier KG, Kendler KS, Ohlsson H, Sundquist K, Sundquist J. Triangulation of evidence on immigration and rates of alcohol use disorder in Sweden: Evidence of acculturation effects. Alcohol (Hanover). 2023; 47(1):104–15. Epub 20221122. https://doi.org/10.1111/acer.14969 PMID: 36336820; PubMed Central PMCID: PMC10016429.
- 43. Grigg D. Wine, Spirits and Beer: World Patterns of Consumption. Geography. 2004; 89(2):99–110.

- Wallace M. Mortality Advantage Reversed: The Causes of Death Driving All-Cause Mortality Differentials Between Immigrants, the Descendants of Immigrants and Ancestral Natives in Sweden, 1997– 2016. Eur J Popul. 2022; 38(5):1213–41. Epub 20221027. https://doi.org/10.1007/s10680-022-09637-0 PMID: 36507238; PubMed Central PMCID: PMC9727037.
- **45.** Westman J, Martelin T, Harkanen T, Koskinen S, Sundquist K. Migration and self-rated health: a comparison between Finns living in Sweden and Finns living in Finland. Scand J Public Health. 2008; 36 (7):698–705. Epub 20080722. https://doi.org/10.1177/1403494808089649 PMID: 18647791.
- Karriker-Jaffe KJ, Ji J, Sundquist J, Kendler KS, Sundquist K. Disparities in pharmacotherapy for alcohol use disorder in the context of universal health care: a Swedish register study. Addiction. 2017; 112 (8):1386–94. Epub 20170516. <u>https://doi.org/10.1111/add.13834</u> PMID: <u>28406579</u>; PubMed Central PMCID: PMC5503767.
- Stockwell T, Zhao J, Panwar S, Roemer A, Naimi T, Chikritzhs T. Do "Moderate" Drinkers Have Reduced Mortality Risk? A Systematic Review and Meta-Analysis of Alcohol Consumption and All-Cause Mortality. J Stud Alcohol Drugs. 2016; 77(2):185–198. https://doi.org/10.15288/jsad.2016.77. 185 PMID: 26997174.
- **48.** Nordic Council of Ministers. Alcohol: Nordic Council of Ministers. 2023 [Accessed 2023/12/19]. Available from: https://pub.norden.org/nord2023-003/alcohol-.html#:~:text=Based%20on%20this%20and% 20new,intake%20should%20be%20very%20low.